Author's response to reviews

Title: Vitalum study design: RCT evaluating the efficacy of tailored print communication and telephone motivational interviewing on multiple health behaviors

Authors:

Hilde M van Keulen (h.vankeulen@gvo.unimaas.nl)
Ilse Mesters (i.mesters@gvo.unimaas.nl)
Johannes Brug (j.brug@vumc.nl)
Marlein Ausems (marlein.ausems@av-m.nl)
Marci Campbell (marci.campbell@unc.edu)
Ken Resnicow (kresnic@umich.edu)
Paul J Zwietering (paul.zwietering@hag.unimaas.nl)
Gerard van Breukelen (gerard.vbreukelen@stat.unimaas.nl)
Willem van Mechelen (w.vanmechelen@vumc.nl)
Johan L Severens (h.severens@beoz.unimaas.nl)
Hein de Vries (hein.devries@gvo.unimaas.nl)

Version: 4 Date: 9 June 2008

Author's response to reviews: see over
Dear Editor,

Herewith, I would like to resubmit the latest version of the study protocol. There were 3 small typographical changes made:

1) Tables – saturated fat intake was used in stead of saturate fat intake;
2) Figure 4 – point 4: behavior(s) in stead of behavior;
3) Figure 4 – point 5 b for not ready: behavior in stead of behaviour.

Please let me know if you have questions or if changes are incorrect.

With kind regards,

Hilde van Keulen

Maastricht University
School for Public Health and Primary Care (Caphri)
Department of Health Education and Health Promotion
P.O. Box 616, 6200 MD Maastricht, the Netherlands
Telephone: + 31 43 3882448
E-mail: h.vankeulen@gvo.unimaas.nl